\begin{table}%T7
\caption{Dwarf novae discovered in the HQS with their
subtype, U~Gem (UG), SU~UMa (SU), Z~Cam (ZC), and unclassified
(XX). Uncertain classifications are marked by a colon.\label{t-hqsdn}}
%\centerline
 {\small
\begin{tabular}{llccc}
\hline\hline\noalign{\smallskip} 
HQS ID & Other name & \Porb~(min) & Type & References\\
\hline\noalign{\smallskip}
%
HS~2331+3905 &         & 81.1  & WZ: & 1 \\
HS~1449+6415 & KV~Dra & 84.9  & SU & 2, 3 \\
HS~2219+1824 &         & 86.3  & SU & 4 \\
HS~1340+1524 &         & 92.7  & XX & 5 \\
HS~1017+5319 & KS~UMa & 97.9  & SU & 2, 6 \\
HS~0417+7445 &         & 105.1/109.9 & SU & 5 \\
HS~1016+3412 &         & 114.3  & XX & 5 \\
HS~0913+0913 & GZ~Cnc & 127.1 & XX & 2, 8, 9 \\
HS~0941+0411 & RX~J0944.5+0357 & 215.0 & XX & 2, 10 \\
HS~0552+6753 & LU~Cam & 216.6 & UG & 2, 7 \\
HS~0907+1902 & GY~Cnc & 252.6 & UG & 2, 11 \\
HS~2214+2845 &         & 258.0 & UG & 5 \\
HS~1857+7127 &         & 272.3 & ZC: & 5 \\
HS~1804+6753 & EX~Dra & 302.3 & UG & 12, 13, 14 \\
%
\noalign{\smallskip}\hline\noalign{\smallskip}
\end{tabular}}
\smallskip
References: (1) \citet{araujo-betancoretal05-1}; (2)
\citet{jiangetal00-1}; (3) \citet{nogamietal00-1}; (4)
\citet{rodriguez-giletal05-1}; (5) this work; (6)
\citet{pattersonetal03-1}; (7) Thorstensen priv. com., \&
vsnet-campaign-dn 2681; (8) \citet{katoetal02-1}; (9)
\citet{tappert+bianchini03-2}; (10) \citet{mennickentetal02-1}; (11)
\citet{gaensickeetal00-2}; (12) \cite{fiedleretal97-1}; (13)
\cite{billingtonetal96-1}; (14) \citet{shafter+holland03-1}.
\end{table}